

## **Author's response to reviews**

**Title:**Laparoscopic versus open adhesiolysis for small bowel obstruction - a multicenter, prospective, randomized, controlled trial

### **Authors:**

Ville Sallinen ([ville.sallinen@helsinki.fi](mailto:ville.sallinen@helsinki.fi))  
Heidi Wikström ([heidi.wikstrom@hus.fi](mailto:heidi.wikstrom@hus.fi))  
Mikael Victorzon ([mikael.victorzon@vshp.fi](mailto:mikael.victorzon@vshp.fi))  
Paulina Salminen ([paulina.salminen@tyks.fi](mailto:paulina.salminen@tyks.fi))  
Vesa Koivukangas ([vesa.koivukangas@ppshp.fi](mailto:vesa.koivukangas@ppshp.fi))  
Eija Haukijärvi ([eija.haukijarvi@pshp.fi](mailto:eija.haukijarvi@pshp.fi))  
Berndt Enholm ([berndt.c.enholm@phsotey.fi](mailto:berndt.c.enholm@phsotey.fi))  
Ari Leppäniemi ([ari.leppaniemi@hus.fi](mailto:ari.leppaniemi@hus.fi))  
Panu Mentula ([panu.mentula@hus.fi](mailto:panu.mentula@hus.fi))

**Version:**3**Date:**25 September 2014

**Author's response to reviews:** see over

Point-by-point response

Additional request formatting:

1. Please include the date of trial registration in the abstract alongside the Trial Registration Number.

**Answer:** Done.

2. Can you please include a completed CONSORT checklist as an additional file when submitting your revised manuscript.

**Answer:** Checklist is included.

Editorial Comments:

1. Ethical approval  
Please provide in your methods section, a full list of all participating centers and the institutional review boards that approved your study in each center.

**Answer:** We have included the centers in the Methods.